Author’s response to reviews


Authors:
Lian Duan (jasonmao2005@163.com)
Rufei Shen (583015777@qq.com)
The Yu Song (songlingyu1991@163.com)
Yong Liao (liaoyong00133@sina.com)
Hongting Zheng (fnf7703@hotmail.com)

Version: 2 Date: 17 Jan 2018

Author’s response to reviews:

Dear Pro. Dickens,

Thank you for carefully reading our case report, we have done the following correction:

1. We have indicated that the mother of the patient gave consent in ‘Consent to Publish statement’.
2. We uploaded again all additional files and figure files alongside the manuscript
3. We added a list of all the abbreviations used in the manuscript before the Declarations section.
4. We resubmitted our manuscript without track changes or highlighting.

We look forward to receiving your email if there is any question.